

Author's response to reviews

Title: Impact of Matrix Metalloproteinase 9 rs3918242 Genetic Variant on Lipid-lowering Efficacy of Simvastatin Therapy in Chinese Patients with Coronary Heart Disease

Authors:

Yuanyuan Xu (snail5213344@126.com)

Yan Wang (wangshen668@126.com)

Jixin Zhi (zhijixin23@163.com)

Lichun Qi (qilichun423@163.com)

Tong Zhang (kinepo@163.com)

Xueqi Li (Xueqili2016@outlook.com)

Version: 3 Date: 06 Feb 2017

Author's response to reviews:

1. We list the Additional file information after the References. We avoid using full sentences to provide information on additional file.
2. We reformulate the large sections of highlighted text present in your Introduction and Methods sections.